Author's response to reviews

Title: SLC6A3 and body mass index in the Prostate, Lung, Colorectal and Ovarian Cancer Screening Trial

Authors:

Elizabeth M Azzato (ema32@cam.ac.uk)
Lindsay M Morton (mortonli@mail.nih.gov)
Andrew W Bergen (andrew.bergen@sri.com)
Sophia S Wang (wangso@mail.nih.gov)
Nilanjan Chatterjee (chattern@mail.nih.gov)
Paul Kvale (pkvale1@hfhs.org)
Meredith Yeager (my76y@nih.gov)
Richard B Hayes (hayesr@mail.nih.gov)
Stephen J Chanock (sc83a@nih.gov)
Neil E Caporaso (caporasn@epndce.nci.nih.gov)

Version: 5 Date: 14 January 2009

Author's response to reviews:

To Whom It May Concern:

Please see our responses to the requested formatting changes below.

1) Please use sentence case in the title.
Our title "SLC6A3 and body mass index in the Prostate, Lung, Colorectal and Ovarian Cancer Screening Trial" is currently in sentence case. The "Prostate, Lung, Colorectal and Ovarian Screening Trial" is a proper noun and requires capitalization.

2) The reference list should contain all author's names.
We have adjusted the references to no longer include "et al." and instead listed all authors names. The exceptions to this are references 7, 8, and 12, which have greater than 30 authors.

Thank you.

Sincerely,
Elizabeth Azzato